Author’s response to reviews

Title: Phytochemical profiling and in vitro screening for anticholinesterase, antioxidant, antiglucosidase and neuroprotective effect of three traditional medicinal plants for Alzheimer's Disease and Diabetes Mellitus dual therapy

Authors:
Amooru Gangaiah Damu (agdamu01@gmail.com;agdamu01@yahoo.com)
Penumala Mohan (mohan.penumala@gmail.com)
Zinka Raveendra Babu (zinkaravi@gmail.com)
Shaik Jeelan Basha (jeelan01chem@gmail.com)
Mallepalli Suresh Kumar Reddy (sureshbiotech2005@gmail.com)
Vadde Ramakrishna (vrkrishna70@gmail.com)

Version: 4 Date: 23 Sep 2017

Author’s response to reviews:

Greetings for today. We thank the reviewers for their valuable suggestions and comments which helped to strengthen our manuscript. According to their suggestions, changes have been incorporated in the revised manuscript. The point by point changes and explanation to the reviewer’s comments are given below.

Answer to the queries raised by Reviewer

1. Email addresses of all authors are included in the title page of manuscript and revised the manuscript.